Author’s response to reviews

Title: Trends in hospital admissions due to antidepressant-related adverse drug events from 2001 to 2011 in the U.S.

Authors:

Harish Parihar (harishparihar@hotmail.com) (harishpa@pcom.edu)
Hongjun Yin (hongjunyin@yahoo.com)
Jennifer Gooch (jennifergooch@pcom.edu)
Shari Allen (sharial@pcom.edu)
Samuel John (samueljo@pcom.edu)
Jianwei Xuan (jwx02467@gmail.com)

Version: 2 Date: 05 Jan 2017

Author’s response to reviews:

1) Please change the title of the Results and Discussion section to Results, and edit the Discussion section below to add Discussion as a header.

Response: Revised as per instructions

2) Declarations

Please ensure that your manuscript contains a Declarations section, with all of the following subsections:

* List of abbreviations
* Ethics approval and consent to participate
* Consent to publish
* Availability of data and materials
* Competing interests
* Funding
* Authors' contributions
* Acknowledgements.

Response: Revised as per instructions.